Author’s response to reviews

Title: Developing a web-based information resource for palliative care: an action-research inspired approach

Authors:

- Annette F Street (A.Street@latrobe.edu.au)
- Kathleen Swift (K.Swift@boroondara.vic.gov.au)
- Merilyn Annells (M.Annells@latrobe.edu.au)
- Roger Woodruff (RogerWoodruff@access.net.au)
- Terry Gliddon (Terry.Gliddon@rdns.com.au)
- Anne Oakley (AOakley@mcm.org.au)
- Goetz Ottmann (goetz.ottmann@gmail.com)

Version: 5 Date: 11 September 2007

Author’s response to reviews:

Dear Editor,

As requested, in this version (v6) the endnote numbers following the affiliations of each authors have been replaced with in text numerals, the affiliation of author no 7 has been corrected, the title 'corresponding author' has been corrected, and the tables have been renamed and reordered.

Best Regards,

Goetz Ottmann